 
CENWP-OP-B                                                                                                                                 8 October, 2001

MEMORANDUM FOR CENWP-OP, ATTN: Gary Johnson

SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for week 40 of 2001, which covers the period 30 September through 06 October.
1.  OPERATION SUMMARY.

1.1 Daily river flow averaged 78.3 to 101.0kcfs.   Daily project tailwater elevation averaged 7.2’ to 9.2’ msl.  Daily project forebay elevation averaged 75.2’ to 75.8’ msl.  River temperature averaged 65 to 66° F.

1.2 Spill bays 1 and 18, which supply attraction flow for Cascades Island and Bradford Island ladders respectively, remain open 4”.  This translates to a daily spill average of 2.3kcfs.

1.3 Unit 3 remains out of service for turbine rehab.  Unit 17 remains out of service for its four-year overhaul.  Unit 5 remains out of service for turbine rehab.   Unit 9 returned to service at 1554 on 02 October.

1.4 Second Powerhouse fish units remain in service.  They provide attraction flow for the Washington Shore fish ladder.  The units are placed on standby from approximately 0000-0300 nightly if drawdown exceeds 1.0’.  

2.  ACTIVITIES. 
2.1 AWS Closures. FV4-3 and FV4-4 at B-branch and FV5-3 and FV5-4 are all operating in manual until the new computer system is calibrated.  FV5-9 was closed for cleaning from 0701 until 0737 on 01 October, and from 1530 until 1545 on 04 October.  FV1-1 was closed for cleaning from 1459 until 1530 on 01 October.  The Bradford Island exit was closed for cleaning from 1510 until 1525 on 04 October and again from 1310 until 1400 on 06 October.  FV3-7 and FV3-9 were closed for cleaning from 1310 until 1400 on 06 October.

2.2 STS/VBS Inspections.  No STSs were inspected this reporting week.

2.3 STS removal. PH1 STS remain OOS per NMFS request.

2.4 Dewatering and Fish Salvage.  On 03 October, Unit 17 taillogs were pulled, there were no fish to salvage.  On 4 October, at 0900, spillbay 17 was dewatered.  Biologists removed 1 dead sturgeon, tried to remove 1 decomposing adult Chinook and a handful of resident sculpin, suckers, etc.   On 4 October, at 1300, DSM1 was dewatered to conserve water.  Biologists removed 6 sculpin, 4 crayfish, 1 bass and 1 unclipped juvenile Steelhead.  All fish were released downstream.

3.  RESEARCH.

3.1 WDFW Fish Counting.  WDFW continues 16-hour visual counts and 8-hour video counts at Bradford Island and Washington Shore ladders.

3.2 NMFS Saltwater rearing. The Saltwater rearing facility is in operation.  Sampling May- November.

3.3 PSMFC Adult PIT.  Tagging completed.  Planning for construction at the Bradford Island and Cascade Island ladders continues.

3.4 PSMFC Smolt Monitoring Program.  Collection continues 24h/d at the PH2 monitoring facility. 

3.5 UI Adult Fallback. UI Adult Radiotracking/Tagging.  PCC entrance gates at the first powerhouse are being operated according to UI request until 31 October.  Tagging of Chinook began at the AFC&MF 2 April and will continue seven days a week until the middle of October.  

3.6 CRITFC/ODFW/WDFW Adult Trapping/Tagging.  CRITFC finished tagging at the end of September.  ODFW and WDFW continue trapping and tagging steelhead on Tuesdays and Thursdays.

3.7 NMFS Lamprey.  Tracking continues.  Operation of the spillway ladders for lamprey test mode (.5’ differential) ended 30 September.

3.8 YIN Northern Pikeminnow Dam Angling.  Fishing 25 June – 30 September.

3.9 WDFW/ODFW Fall Chinook Counting.  Observations of the fall Chinook ended 30 September.

4.  FISHWAYS. 
4.1 Project biologists inspected 30 September through 06 October this reporting week.   

4.2 Fish Passage Plan Observations.

DATE
LOCATion
FPP Deviation
CAUSE
RESPONSE
TR/WAR

All week
PH2 collection channel electronic velocity meter.
The velocity meter is not yet working.
Not fully installed.
Working on final installation.


All week
FG3-28
Out of service
Manually closed because of displaced diffuser valve cover.
New covers are designed, and will be installed during winter maintenance.
20013904

All week
FG3-27
Not in correct position for TW elev.
FG3-27 is open to replace FG3-28.
No action taken.


All week
PH1 Collection channel velocity
Velocity meter registers less than 1.5 fps.
Water backs up in the channel during OG/SG tests.  
Monitoring situation.


All week
Main Dam Avian wires
Wires removed.
Spillway deflector installation.
Wire grid to be replaced in Mid-April.


10/2 
Orifice 13A-S
Closed until winter maintenance season.
Shaft has separated from the knife gate.
Operators put in local control.  
20012261



10/4
B-branch staff gauge
Reading >1.1’
South entrance gate closed
Operators to open gate.


10/4
B-branch entrance gates
Not in correct position for TW elev.
System is still manually operated.
Operators to open gate.


4.3 Fish Ladder Temperatures.  Temperatures are measured 3 weirs down from the count station at Bradford Island and at the junction of the UMT and the main ladder at Washington shore.

Date
Wash. Shore
Bradford Is.

9/30
18.6
18.7

10/01
18.4
18.3

10/02
18.6
18.4

10/03
18.6
18.5

10/04
18.3
18.2

10/05
17.8
17.8

10/06
17.7
17.5

4.4 Sensor Calibration.  Water level sensors were checked for calibration on 05 October.  

4.5 Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week. 

5. MISCELLANEOUS

5.1 PH2 JBS injury/mortality issue: Juvenile mortality remains low at the JMF, according to PSMFC reports.  

5.2 Because of low tailwater elevations and the physical limits of the weir gates, entrance openings for PH1 Collection Channel are less than 8.0’.   Gate 64 has been reprogrammed to maintain at least an 8’ opening.  

5.3 Closed OG/SG testing causes low differential at the north end of PH1 collection channel.  FV1-1 normally controls south channel differential, but is now reprogrammed to maintain north channel differential.  This change should also help maintain proper collection channel velocity during OG/SG tests.  

5.4 The B-Branch and Cascade Island computers were installed.  Sensors need calibrated and some parameters need setting.  Until then, entrance gates and FV4-3, 4-4, 5-3, and 5-4 are operated in manual.  Waiting on parts.

5.5 Weir 37 float control doesn’t adequately control water level over the weir.  A sonic transducer will be installed which should correct the problem.  

5.6 The outfall hydrocannon remains off.  We have not observed any bird activity.

5.7 DSM1 was dewatered on 04 October to conserve water.  

5.8 Spillbay deflector construction requires use of a dewatering bulkhead.  A construction representative informed project biologists that he’d seen fish in the area behind the bulkhead.  We found the area extremely difficult to access and fish rescue complicated.  We made reasonable efforts to remove salmonids and sturgeon, but safety concerns led us to instruct the contractor to move small resident fish to the tailrace by bucket.  Planning division biologists are devising a dewatering plan for the remaining deflectors.

                                                                                                                DEBORAH L. CHENOWETH

                                 

                                                             Operations Project Manager
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